Author’s response to reviews

Title: Combining learning for educators and participants in a paediatric CPD programme

Authors:

Daniel Holmgren (daniel.holmgren@vregion.se)
Maria Skyvell-Nilsson (maria.skyvell-nilsson@hv.se)
Per Wekell (per.wekell@vregion.se)

Version: 3 Date: 08 Jan 2019

Author’s response to reviews:

The Editor

BMC Medical Education

8 January 2019

MEED-D-18-00666

Dear Editor,

On behalf of all the authors, we would like to submit our manuscript, Combining learning for educators and participants in a paediatric CPD programme, by Holmgren D et al., to BMC Medical Education after a minor revision.

The revision has been made according to the recommendations made by the Editor.

We hope that the manuscript now may be considered for publication.

Yours sincerely,

Daniel Holmgren, MD, PhD
Associate Professor

Director of Study for Paediatric CPD in Western Sweden

Departments of Paediatrics Skaraborg Hospital, Skövde and Gothenburg University, Sweden

Telephone +46703618750